Author's response to reviews

Title: "A powerful intervention: GPs' use of sickness certification in depression"

Authors:

Sara M Macdonald (sara.macdonald@glasgow.ac.uk)
Maragaret Maxwell (margaret.maxwell@stir.ac.uk)
Philip Wilson (phil.wilson@glasgow.ac.uk)
Michael Smith (michael.smith3@nhs.net)
William Whittaker (william.whittaker@manchester.ac.uk)
Matt Sutton (matt.sutton@manchester.ac.uk)
Jillian Morrison (jill.morrison@glasgow.ac.uk)

Version: 2 Date: 5 April 2012

Author’s response to reviews:

Dear Editor

As per your email of 5 April 2012, I have carried out the following revisions:

1. Details of the ethical approval for the study now appear in the methods section.
2. The manuscript now contains an acknowledgements section.
3. The authors contributions have been detailed.
4. The funders of the study have been acknowledged.
5. The abstract has been revised to comply with the house style.

Many thanks for your patience,

regards

Sara Macdonald